Author's response to reviews

Title: Prevalence of vancomycin-resistant Enterococcus fecal colonization among kidney transplant patients

Authors:

   Maria Cecilia S Freitas (mfreitas@mednet.ucla.edu)
   Alvaro Pacheco-Silva (apacheco@nefro.epm.br)
   Dulce Barbosa (dulce@denf.epm.br)
   Suzane Silbert (suzanesilbaert@terra.com.br)
   Helio Sader (helio-sader@jmilabs.com)
   Ricardo Sesso (rsesso@nefro.epm.br)
   Luis Fernando A Camargo (luisfacamargo@uol.com.br)

Version: 3  Date: 15 August 2006

Author's response to reviews:

Dear editor,

We did all the formatting changes that were requested:

1. Title: All unnecessary capitals were removed;
2. Author list: The author list is in the same order which they are listed in the submission system;
3. Each affiliation was written only once and e-mail addresses were written separately from the affiliations;
4. Section headings: All unnecessary capitols were removed;
5. Competing interests: OK;
6. Tables: Were moved to a position after the references;
7. References: Contains all authors names and 'et al' was not used.

Sincerely,

Maria Cecilia de Santos Freitas